Author's response to reviews

Title: A highly efficacious pediculicide based on dimeticone: randomized observer blinded comparative trial in patients with severe infestation

Authors:

Jorg Heukelbach (heukelbach@web.de)
Daniel Pilger (daniel.pilger@web.de)
Fabiola A Oliveira (foliveira@web.de)
Adak Khakban (adak_khakban@yahoo.com)
Liana Ariza (arizaliana@hotmail.com)
Hermann Feldmeier (hermann.feldmeier@charite.de)

Version: 2 Date: 12 May 2008

Author's response to reviews:

Dear Sir or Madam:
Please find enclosed the revised manuscript including the ISRCTN number.
Thank you very much for collaboration.
Yours sincerely,
Jorg Heukelbach